Author’s response to reviews

Title: Intrathecal blood injection: A case report of a rare complication of an epidural blood patch

Authors:

Joseph Seemiller (joe@josephseemiller.com)
Sankeerth Challagundla (schallagundla@geisinger.edu)
Travis Taylor (ttaylor1@geisinger.edu)
Ramin Zand (rzand@geisinger.edu)

Version: 1 Date: 24 Mar 2020

Author’s response to reviews:

Dear Editor,

I have made the requested changes in this revised version including:

(1) Updated contact information for the corresponding author

(2) Clarified role of the funder

(3) Contributions heading is named "Author's Contributions"

(4) Figure legends moved to a separate heading at the end of the document

Thank you for your help in editing our manuscript.